Author's response to reviews

Title: Measuring factors affecting implementation of health innovations: A systematic review of structural, organizational, provider, patient, and innovation level measures

Authors:

Stephenie R. Chaudoir (schaudoir@gmail.com)
Alicia G. Dugan (alia.dugan@uconn.edu)
Colin H. I. Barr (colin.barr@uconn.edu)

Version: 8 Date: 31 January 2013

Author's response to reviews: see over
Hello,
Per your request, I have accepted all of your changes to the copy edited manuscript.

Please be sure to note that:
1. Stephenie Chaudoir should be listed as the corresponding author.
2. I rejected your change on pg. 11, paragraph 1. The second use of the phrase ‘research utilization’ should be spelled as ‘research utilisation’ with an ‘S’ to reflect the UK spelling; it was used as a separate keyword in our search.

Thank you,
Stephenie Chaudoir